Author's response to reviews

Title: Patterns of active and passive smoking, and associated factors in the Southeast Anatolian Project (SEAP) region, in Turkey

Authors:

Ali Ihsan Bozkurt (abozkurtpamukkale.edu.tr)
Saime Sahinoz (saimemail.com)
Birgul Ozcirpici (ozcirpici@gantep.edu.tr)
Servet Ozgur (ozgur@gantep.edu.tr)
Turgut Sahinoz (turgut29@mynet.com)
Hamit Acemoglu (hacemoglu@dicle.edu.tr)
Gunay Saka (gsaka@dicle.edu.tr)
Ali Ceylan (alic@dicle.edu.tr)
Yilmaz Palanci (yilmazp@dicle.edu.tr)
Erser Ilcin (ilcine@dicle.edu.tr)
Feridun Akkafa (aferidun@harran.edu.tr)

Version: 7 Date: 25 January 2006

Author's response to reviews:

Dear Editor,

Please find attached a revised paper (final version) named "Patterns Of Active and Passive Smoking And Associated Factors in South-east Anatolian Project (SEAP) Region, Turkey".
Thank you for the opportunity to submit our work. Yours truly,

Ali Ihsan Bozkurt, MD, Ph.D.

Running title: removed from the title page
Title: removed all unnecessary capitals
Authors' contributions: this section included between the competing interests and Acknowledgements sections

The Department of Public Health, Faculty of Medicine, Pamukkale University, Kinikli, Denizli, TURKEY
e-mail: abozkurtpamukkale.edu.tr
Tel: +90 258 2138177-2138178, Fax: +90 258 2132874